Author's response to reviews

Title: Endurance, resistance and resilience in the South African health care system: a comparative case study to demonstrate mechanisms of coping within a transforming system

Authors:

John Eyles (john.eyles@wits.ac.za)
Bronwyn Harris (bronwyn.harris@wits.ac.za)
Jana Fried (jana.fried@usask.ca)
Veloshnee Govender (veloshnee.govender@uct.ac.za)
Pascalia Munyewende (pascalia.munyewende@wits.ac.za)

Version: 2 Date: 20 November 2014

Author's response to reviews:

Dear Editors

The line and page numbers have now been added to the document. We would like to apologise for our oversight.

Sincerely,
The Authors